\begin{thebibliography}{10}\setlength{\itemsep}{-1ex}\small

\bibitem{Schmidt:05o}
K.~Balasubramanian, J.~Balasubramanian, J.~Parsons, A.~Gokhale, and D.~C.
  Schmidt.
\newblock {A Platform-Independent Component Modeling Language for Distributed
  Real-time and Embedded Systems}.
\newblock {\em {Elsevier Journal of Computer and System Sciences}}, 2006.
\newblock To Appear.

\bibitem{Generative:Programming:00}
K.~Czarnecki and U.~W. Eisenecker.
\newblock {\em {Generative Programming: Methods, Tools, and Applications}}.
\newblock Addison-Wesley, Reading, Massachusetts, 2000.

\bibitem{Schmidt:03n}
A.~Gokhale, D.~C. Schmidt, B.~Natarajan, J.~Gray, and N.~Wang.
\newblock {Model Driven Middleware}.
\newblock In Q.~Mahmoud, editor, {\em Middleware for Communications}, pages
  163--187. Wiley and Sons, New York, 2004.

\bibitem{Janos:03}
G.~Karsai, J.~Sztipanovits, A.~Ledeczi, and T.~Bapty.
\newblock {Model-Integrated Development of Embedded Software}.
\newblock {\em Proceedings of the IEEE}, 91(1):145--164, Jan. 2003.

\bibitem{GME:01}
A.~Ledeczi, A.~Bakay, M.~Maroti, P.~Volgysei, G.~Nordstrom, J.~Sprinkle, and
  G.~Karsai.
\newblock {Composing Domain-Specific Design Environments}.
\newblock {\em IEEE Computer}, pages 44--51, November 2001.

\bibitem{CORBAService:99c}
{Object Management Group}.
\newblock {\em {Fault Tolerant CORBA Specification}}, {OMG Document
  orbos/99-12-08} edition, Dec. 1999.

\bibitem{DandC:03}
{Object Management Group}.
\newblock {\em {Deployment and Configuration Adopted Submission}}, {OMG
  Document mars/03-05-08} edition, July 2003.

\bibitem{LWCCM-2004}
{Object Management Group}.
\newblock {\em {Lightweight CCM FTF Convenience Document}}, {ptc/04-06-10}
  edition, June 2004.

\bibitem{PFPN:IeeeTocs}
{Pascal Felber and Priya Narasimhan}.
\newblock {Experiences, Approaches and Challenges in building Fault-tolerant
  CORBA Systems}.
\newblock {\em {Transactions of Computers}}, 54(5):497--511, May 2004.

\bibitem{Mead:CCPE}
{Priya Narasimhan and Tudor Dumitras and Aaron M. Paulos and Soila M. Pertet
  and Charlie F. Reverte and Joseph G. Slember and Deepti Srivastava}.
\newblock {MEAD: support for Real-Time Fault-Tolerant CORBA.}
\newblock {\em {Concurrency - Practice and Experience}}, 17(12):1527--1545,
  2005.

\bibitem{Schmidt:06a}
D.~C. Schmidt.
\newblock {Model-Driven Engineering}.
\newblock {\em IEEE Computer}, 39(2):41--47, 2006.

\end{thebibliography}

% $Id: overview.tex 5108 2007-01-07 23:52:29Z gokhale $

\subsection{Overview of Enabling Technologies}
\label{sec:overview}

Before delving into the details of our design-time dependability
provisioning tool, we first provide an overview of the enabling
technologies we have leveraged to develop MDDPro.

MDDPro has been developed in the context of the CoSMIC (Component
Synthesis with Model Integrated Computing)~\cite{Schmidt:03n} MDE
toolsuite.  CoSMIC is an open source MDE tool suite used to simplify
the development of component-based DRE applications focusing
particularly on the assembly, deployment, configuration, and
validation of component-based enterprise DRE systems.  CoSMIC
comprises a collection of {\em domain-specific modeling languages}
(DSMLs), which define the concepts, relationships, and constraints
used to express domain entities~\cite{Janos:03}, and generative
programming capabilities that automate the different development
concerns of DRE systems.

The different capabilities in CoSMIC including the MDDPro tool
described in this paper have been developed using the Generic Modeling
Environment (GME)~\cite{GME:01}.  GME is a metaprogrammable modeling
environment that enables domain experts to develop visual modeling
languages and generative tools associated with those languages.  The
modeling languages in GME are represented as metamodels.  A metamodel
in GME depicts a class diagram using UML-like constructs showcasing
the elements of the modeling language and how they are associated with
each other.

A key CoSMIC DSML developed in GME is the {\em Platform Independent
  Component Modeling Language} (PICML)~\cite{Schmidt:05o}, which
enables graphical manipulation of modeling elements, such as component
ports and attributes.  PICML also performs various types of generative
actions, such as synthesizing XML-based deployment plan descriptors
defined in the OMG Deployment and Configuration (D\&C)
specification~\cite{DandC:03}.  CoSMIC provides the {\em
  Component QoS Modeling Language} (CQML), which is a mappping of the
platform-independent PICML models to models that are specific to the
lightweight CORBA Component Model.  Figure~\ref{fig:cqml-sample}
illustrates the CQML model for the enterprise DRE system case study
from Figure~\ref{fig:sample}.  Our MDDPro tool is an enhancement to
the CQML DSML and its generative capabilities.


\begin{figure}[htb]
  \centering
  \includegraphics[width=0.9\columnwidth,keepaspectratio]{figs/slice-cqml}
  \vspace{-3.3in}
  \caption{CQML Model of the Enterprise DRE System Case Study}
  \label{fig:cqml-sample}
\end{figure}



%%% Local Variables: 
%%% mode: latex
%%% TeX-master: "isas07"
%%% End: 
